⋆ Confidential ⋆

Student Evaluation of Presentation

Presenter: Topic: Date:

(20%) Presentation Skills

• Was the presentation visually clear? • Was the presentation audibly clear? • Did the presentation engage your attention? • Did the presenter talk directly to the audience? • Was the speaker confident and familiar with the article? ▶ 0 weak

1 2 3 4 strong

(30%) Content

• Was the presentation insightful? • Were appropriate examples used?

• Did you learn anything from the presentation?

• Do you understand something about the article (or topic) better because of presentation? • Did the audience benefit from the analysis of the article, or of the consequences of the article? ▶ 0 weak

1 2 3 4 strong

(30%) Organization/Structure

• Was the specific purpose of the presentation made clear?

• Was the organization balanced so there was enough time for important concepts? • Was the organization balanced so there was enough explanation of difficult concepts? • Were the ideas presented in a logical sequence? • Was there a clear introduction & conclusion? ▶ 0 weak

1 2 3 4 strong

(20%) Conciseness

• Was the talk too short (< 10 min.) or too long (> 20 min.)? • Were the ideas expressed without rambling? ▶ 0 weak

1 2 3 4 strong

Comments □ I have made more comments on the other side of this page

Your name:

⋆ This is confidential ⋆ CS 6606 (Fall 2001)